Author's response to reviews

Title: The School Anxiety Scale-Teacher Report (SAS-TR): translation and psychometric properties of the Iranian version

Authors:

Zahra Hajiamini (z_hajiamini@hotmail.com)
Ashraf Mohamadi (maryam.mohammadi14@yahoo.com)
Abbas Ebadi (ebadi1347@bmsu.ac.ir)
Ali Fathi Ashtiani (afa1337@gmail.com)
Mahmoud Tavousi (tavousi@acecr.ac.ir)
Ali Montazeri (montazeri@acecr.ac.ir)

Version: 7 Date: 1 July 2012

Author's response to reviews: see over
Dear Deesha,

MS: 8597469905125213
The School Anxiety Scale-Teacher Report (SAS-TR): translation and psychometric properties of the Iranian version

Thank you for your e-mail. Please find the following point-by-point responses as requested:

Reviewer: Dr. Heidi Lyneham

Major Compulsory Revisions
Nil
Thank you.

Minor Essential Revisions
The paper needs to be edited in many places for grammar and spelling. For example in the abstract the line "A sample of students was participated in the study" should be "A sample of students participated in the study". Thank you. This was corrected.
In the introduction "That high rate for co-morbid depression and anxiety ranging from 15.9 to 61.9% has also been documented" should read "A high rate of co-morbid depression and anxiety ranging from 15.9 to 61.9% has also been documented". Thank you. This was corrected.
Again I would encourage the authors to seek assistance from a native English speaking researcher to correct these minor errors. Thank you. The manuscript was further checked for English. Hope you find it satisfactory. The changes are indicated by red.

Discretionary Revisions
Nil
Thank you.

I hope you find the corrections satisfactory.
I wish you all the best.
Kind regards
Ali Montazeri